Author's response to reviews

Title: Seasonal pattern of peptic ulcer hospitalizations: analysis of the hospital discharge data of the Emilia-Romagna region of Italy.

Authors:

Roberto Manfredini (r.manfredini@ausl.fe.it)
Roberto De Giorgio (deg@aosp.bo.it)
Michael H Smolensky (msmolensky@austin.rr.com)
Benedetta Boari (brobdt@unife.it)
Raffaella Salmi (raffaella.salmi@ospfe.it)
Davide Fabbri (davide.fabbri@ospfe.it)
Edgardo Contato (e.contato@ausl.fe.it)
Mauro Serra (mauroserra@email.it)
Giovanni Barbara (giovanni.barbara@unibo.it)
Vincenzo Stanghellini (vincenzo.stanghellini@unibo.it)
Roberto Corinaldesi (robecori@med.unibo.it)
Massimo Gallerani (m.gallerani@ospfe.it)

Version: 4 Date: 9 April 2010

Author's response to reviews:

BMC Gastroenterology, MS 4598922153074184-R
Roberto Manfredini et al.
Seasonal pattern of peptic ulcer hospitalizations: analysis of the hospital discharge data of the Emilia-Romagna region of Italy.

Dear Sir,
Thank you very much for your kind message dated April 8.
I am pleased to enclose the revised version modified accordingly. In particular:

Major revisions
- Figures have been renumbered (1-2-3-4 instead of 1-2-3A-3B). A short title has been provided. The format has been changed for enhancing quality (the actual format is .jpg)
- Tables have been fitted into the suggested dimension (max 176 x 230 mm). They have been checked for accuracy in referencing in the manuscript, and Table 3 is now appropriately cited.

Minor revisions
- Title page: ‘title page’ removed
Should you have further queries or additional changes are to be made, please do not hesitate to contact me.

Best regards

Roberto Manfredini

Department of Internal Medicine, Hospital of the Delta, via Valle Oppio 2, Lagosanto, Ferrara, Italy, and Department of Clinical and Experimental Medicine, Section Clinica Medica, University of Ferrara, via Savonarola 9, Ferrara, Italy.
Tel: +39 0533 723485 Fax: +39 0533 723477
E-mail: r.manfredini@ausl.fe.it - roberto.manfredini@unife.it